Author's response to reviews

Title: Piloting the role of a pharmacist in a community palliative care multidisciplinary team: an Australian experience

Authors:

Safeera Y Hussainy (safeera.hussainy@monash.edu)
Margaret Box (margaretbox@fastmail.fm)
Sandy Scholes (scholes2@bigpond.net.au)

Version: 6 Date: 27 October 2011

Author's response to reviews:

Dear Sir/Madam,

The proof has been revised in the following manner:

- Corresponding author now correctly indicated
- The number 52 is now spelt out in full at the beginning of the first sentence in the 2nd paragraph on page 20

Kind regards
Safeera Hussainy.